Awareness and use of evidence-based medicine information among patients in Croatia: a nation-wide cross-sectional study.

Nejašmić, Danijel; Miošić, Ivana; Vrdoljak, Davorka; Permozer, Hajdarović; Tomičić, Marion; Gmajnić, Rudika; Diminić Lisica, Ines; Sironić, Hreljanović; Pleh, Vlatka; Cerovečki, Venija; ...

Source / Izvornik: Croatian medical journal, 2017, 58, 300 - 309

Journal article, Published version Rad u časopisu, Objavljena verzija rada (izdavačev PDF)

Permanent link / Trajna poveznica: https://urn.nsk.hr/urn:nbn:hr:184:294865

Rights / Prava: Attribution 3.0 Unported / Imenovanje 3.0

Download date / Datum preuzimanja: 2025-01-11



Repository / Repozitorij:

<u>Repository of the University of Rijeka, Faculty of</u> Medicine - FMRI Repository





300

Awareness and use of evidence-based medicine information among patients in Croatia: a nation-wide crosssectional study

Aim To determine the use of evidence-based medicine (EBM) information and the level of awareness and knowledge of EBM among patients in Croatia.

Methods A cross-sectional study was conducted among 987 patients in 10 family medicine practices in Croatia. Patients from both urban (n=496) and rural (n=482) areas were surveyed. A 27-item questionnaire was used to collect data about sources that patients searched for medical information, patient awareness and use of Cochrane systematic reviews and other EBM resources, and their demographic characteristics.

Results Half of the patients searched for medical information from sources other than physician. Internet was the most common place they searched for information. Very few patients indicated using EBM sources for medical information; one fifth of patients heard of EBM and 4% of the patients heard of the Cochrane Collaboration. Patients considered physician's opinion as the most reliable source of medical information. A logistic regression model showed that educational level and urban *vs* rural residence were the predictors of awareness about EBM and systematic reviews (P < 0.001 for both).

Conclusion Our finding that patients consider a physician's opinion to be the most reliable source of health-related information could be used for promotion of high-quality health information among patients. More effort should be devoted to the education of patients in rural areas and those with less formal education. New avenues for knowledge translation and dissemination of high-quality health information among patients are necessary.

Danijel Nejašmić¹, Ivana Miošić², Davorka Vrdoljak³, Snježana Permozer Hajdarović⁴, Marion Tomičić³, Rudika Gmajnić⁵, Ines Diminić Lisica⁶, Jelena Sironić Hreljanović⁷, Vlatka Pleh⁸, Venija Cerovečki⁹, Anita Tomljenović¹⁰, Sanja Bekić¹¹, Minka Jerčić¹², Karla Tuđa¹², Livia Puljak²

¹Department of Medical Physics and Biophysics, University of Split School of Medicine, Split, Croatia

²Cochrane Croatia, University of Split School of Medicine, Split, Croatia

³Department of Family Medicine, University of Split School of Medicine, Split, Croatia

⁴Family Medicine Practice, Kotoriba, Croatia

⁵Family Medicine Practice, Osijek, Croatia

⁶Family Medicine Practice, Vežica, Rijeka, Croatia

⁷Family Medicine Practice, Viškovo, Rijeka, Croatia

⁸Family Medicine Practice, Čakovec, Croatia

⁹Department of Family Medicine, Andrija Štampar School of Public Health, University of Zagreb, School of Medicine, Zagreb, Croatia

¹⁰Health center Zagreb-Centar, Family Medicine Practice, Kupinečki Kraljevec, Croatia

¹¹Family Medicine Practice, Višnjevac, Croatia

¹²Family Medicine Practice, Donji Muć, Croatia

> Received: February 17, 2016 Accepted: July 20, 2017

Correspondence to:

Danijel Nejašmić Department of Medical Physics and Biophysics University of Split School of Medicine Šoltanska 2, 21000 Split, Croatia *danijel.nejasmic@mefst.hr*

301

Healthcare service provides advice, examinations, interventions, surgery, and any other form of medical treatment that is legally approved and in accordance with medical methods and principles (1). Considering that the aim of medical care is to benefit patients, then patients are the best persons who can assess the utility of health service (2). A "person-centered care" is defined as "providing care that is respectful of and responsive to individual patient preferences, needs, and values, and ensuring that patient values guide all clinical decisions" (3). This represents a shift from the paternalistic approach of health care professionals to partnership (4). Although this ideal is rarely accomplished, patients have the right to understand their illness, treatment options, and prognosis regardless of whether they decide to participate in clinical decisions (5,6).

With the increasing availability of the Internet, more and more patients search for available information, which requires that such information be accurate, unbiased, and understandable (7). A survey of 1002 consumers found that 57% of respondents turn to the Internet for information that answers their personal medical questions (8). Furthermore, international studies showed that 72% of Internet users in the USA and 71% in Europe conduct healthrelated searches (9,10). This shows that there is a shift in the role of patient from a passive recipient to an active consumer of health information.

Growing awareness for synthesizing numerous evidence that is used in medicine has led to the development of evidence-based medicine (EBM), the purpose of which is to integrate the best available clinical evidence, clinical expertise, and patient values, needs, and desires (11).

EBM is characterized by the synthesis of primary studies into systematic reviews. There are hundreds of thousands systematic reviews available today (12), but the current gold standard in EBM are Cochrane systematic reviews (13). To make Cochrane evidence available to patients, each Cochrane systematic review contains a plain language summary (PLS) written in short format and simple lay language.

In 2008, Cochrane Croatia was founded at the School of Medicine in Split, Croatia (14), with the main goal to promote EBM and systematic reviews among medical students, health care providers, and general population (15). One of the most popular activities of Cochrane Croatia is translation of PLSs to Croatian language. These PLSs are freely available to the public online and via social media (16). Considering increasing availability of Cochrane evidence on interventions and diagnostic accuracy in Croatian language, it would be useful to assess the effect of the Cochrane Croatia knowledge translation and dissemination activities among patients. Having conducted a national survey on the awareness and knowledge of EBM among physicians (17), this study aimed to determine the use of EBM information and awareness and knowledge of EBM among patients.

METHODS

Sample

In this cross-sectional study, we aimed to include a sample of 1000 general practice (GP) patients from Croatia. From each of the 5 Croatian regions (central Croatia, northwestern Croatia, eastern Croatia, north Adriatic and Lika, middle and south Adriatic), one GP practice was selected from urban and one from rural part of the region. Urban GP practices were located in the cities of Čakovec, Osijek, Rijeka, Split, and Zagreb. Rural GP practices were located in Kotoriba, Kupinečki Kraljevec, Donji Muć, Viškovo, and Vuka.

Participants

The aim was to include 100 patients per GP practice in each region. The participants received instructions to fill out the questionnaire in the waiting room, in a private area. The participants were invited consecutively until there were 50 respondents who were women and 50 who were men, to ensure gender balance. If a patient refused to participate, the recruitment process continued on to the next patient until the target number of 100 participants was reached.

The inclusion criteria were \geq 18 years of age. Exclusion criteria were underage, cognitive disorders, and mental health disorders that could preclude proper understanding of the questionnaire.

After reading the written information about the study, patients signed a written informed consent and completed an anonymous questionnaire about the awareness and use of EBM information sources.

Questionnaire

For the purpose of this study, we developed a 27-item questionnaire with open-ended and multiple-choice questions. The questionnaire was piloted among 10

researchers affiliated with Cochrane Croatia, 10 students (who were not exposed to EBM principles), and 10 lay persons to ensure that the wording was appropriate and that the questionnaire takes into consideration appropriateness of both the content and the language. During this, we kept in mind that the questionnaire would be self-administered by lay persons and particular attention was paid to developing plain language. Feedback from the pilot testing was incorporated (changes made on the questions 1, 2, 5, and 25) and once again the questionnaire was pilot-tested on these individuals to ensure content-related, criterion-related, and construct validity of the questionnaire.

Patients were asked about the frequency of visiting physicians, searching for additional information about their therapy and/or diagnosis, Internet use, using medical information sources, their knowledge of EBM, systematic reviews, The Cochrane Library, Cochrane Croatia, and if they were reading Cochrane PLSs translated to Croatian. Their demographic data were also collected, including professional status, education level, sex, and age. Patients also evaluated the reliability of medical information sources and commented on their experience in case they discussed the medical information they found on their own

TABLE 1. General	characteristics	of study	population
------------------	-----------------	----------	------------

with their physician. English translation of the questionnaire is provided as supplementary material (Supplementary questionnaire). The survey was conducted between September 2014 and September 2015.

Statistical analysis

Each guestionnaire received a unique code and every answer was coded with numerical indicators. Data were entered into a piloted data extraction sheet. Descriptive data analysis was performed. Data were expressed as frequencies with percentages or as medians with interquartile ranges (IQR) because the data were not normally distributed, as shown by Kolmogorov-Smirnov test. Comparison of various categorical variables for possible differences between rural and urban areas or regions was conducted using x² test. Association between EBM or systematic review awareness with residence, sex, education level, employment status, visits to physician, and search for health information from sources other than a physician were also tested with χ^2 test. The relation between awareness of EBM or systematic reviews and age was tested using Mann-Whitney test. Logistic regression was used to determine the contribution of several variables

	No. (%) of participants							
Variable	Entire sample (N=978)	Rural areas (n=482)	Urban areas (n=496)	Osijek region (n=189)	Rijeka region (n = 193)	Split region (n=200)	Varaždin region (n=199)	Zagreb region (n=197)
Age (years, median, interquartile range)	47 (34-59)	44 (33-56)	48 (35-62)	45 (35-54)	51 (35-67)	48 (36-62.25)	41 (29-52)	50 (35-62)
Sex (total)	969	479	490	189	193	197	198	192
men	481 (49.6)	235 (49.1)	246 (50.2)	92 (48.7)	96 (49.7)	98 (49.7)	100 (50.5)	95 (49.5)
women	488 (50.4)	244 (50.9)	244 (49.8)	97 (51.3)	97 (50.3)	99 (50.3)	98 (49.5)	97 (50.5)
Employment status (total)	967	479*	488*	187	193*	195	197*	195
employed	489 (50.6)	258 (53.9)	231 (47.3)	108 (57.7)	75 (38.9)	95 (48.7)	114 (57.9)	97 (49.7)
unemployed	160 (16.5)	95 (19.8)	65 (13.3)	40 (21.4)	32 (16.6)	29 (14.9)	34 (17.3)	25 (12.8)
student	66 (6.8)	18 (3.8)	48 (9.9)	5 (2.7)	12 (6.2)	13 (6.7)	22 (11.1)	14 (7.2)
retired	252 (26.1)	108 (22.5)	144 (29.5)	34 (18.2)	74 (38.3)	58 (29.7)	27 (13.7)	59 (30.3)
Education level (total)	966	478*	488*	187	193	195	198	193
primary or lower	131 (13.6)	89 (18.6)	42 (8.6)	25 (13.4)	25 (13.0)	27 (13.8)	32 (16.2)	22 (11.4)
secondary	603 (62.4)	296 (61.9)	307 (62.9)	126 (67.4)	112 (58.0)	123 (63.1)	124 (62.6)	118 (61.1)
college/university	216 (22.4)	85 (17.8)	131 (26.9)	32 (17.1)	51 (26.4)	44 (22.6)	41 (20.7)	48 (24.9)
Master or PhD	16 (1.6)	8 (1.7)	8 (1.6)	4 (2.1)	5 (2.6)	1 (0.5)	1 (0.5)	5 (2.6)
Frequency of visiting a physician/year								
never	29	20	9	12	8	4	4	1
1-2 times	271	144	127	56	57	47	56	55
3-4 times	311	159	152	46	55	75	68	67
6-10 times	205	94	111	39	32	46	39	49
>10 times	156	64	92	36	40	25	30	25
× 3 · · · / 8 · • • • •								

 $^{*}\chi^{2}$ test (*P* < 0.05).

to EBM or systematic review awareness. Analyses were conducted with MedCalc statistical software, v 15.2.1 (© MedCalc Software bvba, Ostend, Belgium). The level of statistical significance was set at P < 0.05.

RESULTS

A total of 978 patients were included in the study. Their median age was 47 years (IQR, 34-59 years). Most patients were employed and had high school-level education. Patients in urban and rural areas had very similar employment and education status in all five geographic regions (Table 1). The majority of patients in both urban and rural areas visited a physician 3-4 times a year, but in two regions the majority of patients indicated visiting physician 1-2 times a year (Table 1).

Half of the patients indicated that they engage in search for medical information on diagnosis or therapy in other

places, ie, they do not only ask their physician. Their most common source of medical information was the Internet, followed by 'friends and acquaintances who are health care workers'. As many as 77% of the patients used Internet, with almost identical percentage of Internet use in urban and rural areas (79% vs 76%, respectively; Table 2). Among 739 patients who indicated using Internet when searching for medical information, there were 658 (89%) who searched for medical information on Internet search engines, such as Google, while a very few patients indicated using PubMed or EBM databases (Table 2).

About one-fifth (19%) of patients indicated they had heard of EBM (Table 3). More patients in urban than rural areas answered positively to this question (25% vs 13%, respectively). The percentages of patients who reported to have heard of systematic reviews, The Cochrane Collaboration, a Cochrane systematic review, Cochrane summary or Cochrane translation were low. Among a few patients who

TABLE 2. Participants' search for medical information

	No. (%) of participants							
Variables	Entire sample (N=978)	Rural areas (n=482)	Urban areas (n=496)	Osijek region (n=189)	Rijeka region (n=193)	Split region (n=200)	Varaždin region (n=199)	Zagreb region (n=197)
Searching for information about diagnosis or therapy other than with physician?	954	473*	481*	185	188	198	197	186
yes	491 (51.5)	224 (47.4)	267 (55.5)	98 (53.0)	92 (49.0)	111 (56.1)	90 (45.7)	100 (53.8)
no	463 (48.5)	249 (52.6)	241 (44.5)	87 (47.0)	96 (51.0)	87 (43.9)	107 (54.3)	86 (46.2)
Sources of medical information ⁺	776	366	410	132	165	161	179	139
books	181 (23.3)	77 (21.0)	104 (25.4)	31 (23.5)	42 (25.5)	20 (12.4)	47 (26.3)	41 (29.5)
friends and acquaintances that are health care workers	343 (44.2)	162 (44.3)	181 (44.1)	62 (47.0)	65 (39.4)	92 (57.1)	71 (39.7)	53 (38.1)
research manuscripts	97 (12.5)	39 (10.7)	58 (14.1)	17 (12.9)	25 (15.2)	7 (4.3)	25 (14.0)	23 (16.5)
promotional materials of pharmaceutical companies	90 (11.6)	42 (11.5)	48 (11.7)	5 (3.8)	29 (17.6)	14 (8.7)	19 (10.6)	23 (16.5)
Internet	552 (71.1)	253 (69.1)	299 (72.9)	93 (70.5)	101 (61.2)	117 (72.7)	127 (70.9)	114 (82.0)
specialized magazines about medicine for patients	105 (13.5)	38 (10.4)	67 (16.3)	8 (6.1)	32 (19.4)	16 (9.9)	9 (5.0)	40 (28.8)
other	41 (5.3)	20 (5.5)	21 (5.1)	3 (2.3)	21 (12.7)	14 (8.7)	3 (1.7)	0 (0.0)
Do you use Internet?	958	465*	493*	189	191	197	197*	184
never	219 (22.9)	99 (21.3)	120 (24.3)	46 (24.3)	58 (30.4)	41 (20.8)	36 (18.3)	38 (20.7)
yes, few times a month	94 (9.8)	59 (12.7)	35 (7.1)	21 (11.1)	24 (12.6)	15 (7.6)	20 (10.1)	14 (7.6)
yes, few times a week	187 (19.5)	108 (23.2)	79 (16.0)	48 (25.4)	34 (17.8)	41 (20.8)	24 (12.2)	40 (21.7)
yes, daily	458 (47.8)	199 (42.8)	259 (52.5)	74 (39.2)	75 (39.2)	100 (50.8)	117 (59.4)	92 (50.0)
If you are using Internet for searching information about the medicine, which Internet information source do you use? [†]	739	366	373	143	133	156	161	146
Internet search engines (Google, etc.)	658 (89.0)	340 (92.9)	318 (85.3)	121 (84.6)	119 (89.5)	131 (84.0)	146 (90.7)	141 (96.6)
PubMed	23 (3.1)	7 (1.9)	16 (4.3)	4 (2.8)	2 (1.5)	6 (3.8)	7 (4.3)	4 (2.7)
specialized evidence-based medicine databases	37 (5.0)	19 (5.2)	18 (4.8)	7 (4.9)	8 (6.0)	8 (5.1)	12 (7.5)	2 (1.4)
other	20 (2.7)	5 (1.4)	15 (4.0)	2 (1.4)	6 (4.5)	1 (0.6)	6 (3.7)	5 (3.4)
* χ^2 test ($P < 0.05$) †Multiple choice question.								

TABLE 3. Use of evidence-based medicine information sources

				No. (%) of	participan	ts		
Variables	Entire sample	Rural areas	Urban areas	Osijek region	Rijeka region	Split region	Varaždin region	Zagreb region
	(11 - 970)	(11 - 402)	(11-490)	(II – 109) 100*	105	(11-200)	(11 - 199)	(11-197)
medicine?	950	4/3	400	109	201	192	194	190
yes	183 (19.1)	60 (12.7)	123 (25.5)	59 (31.2)	32 (17.3)	25 (13.0)	39 (20.1)	28 (14.2)
no	773 (80.6)	413 (87.3)	360 (74.5)	130 (68.8)	153 (82.7)	167 (87.0)	155 (79.9)	168 (85.8)
Did you ever hear about a type of research called a "systematic review"?	960	475	485	188	187	197	192*	196
yes	167 (17.4)	91 (19.2)	76 (15.7)	29 (15.4)	31 (16.6)	33 (16.8)	48 (25.0)	26 (13.3)
no	793 (82.6)	384 (80.8)	409 (84.3)	159 (84.6)	156 (83.4)	164 (83.2)	144 (75.0)	170 (86.7)
Did you ever hear about The Cochrane Collaboration?	966	477	489	189	191	197	194	195
yes	42 (4.3)	17 (3.6)	25 (5.1)	6 (3.2)	8 (4.2)	8 (4.1)	12 (6.2)	8 (4.1)
no	924 (95.7)	460 (96.4)	464 (94.9)	183 (96.8)	183 (95.8)	189 (95.9)	182 (93.8)	187 (95.9)
Did you ever hear about Cochrane systematic review or a Cochrane summary, a Cochrane translation?	949	470	479	188	184	196	187	194
yes	27 (2.9)	10 (2.1)	17 (3.5)	2 (1.0)	4 (2.2)	7 (3.6)	10 (5.3)	4 (2.1)
no	922 (97.1)	460 (97.9)	462 (96.5)	186 (99.0)	180 (97.8)	189 (96.4)	177 (94.7)	190 (97.9)
If yes, where did you hear about it? ⁺	25	8	17	2	3	7	9	4
The Cochrane Library	4 (16.0)	1 (12.5)	3 (17.6)	0 (0.0)	0 (0.0)	2 (28.6)	2 (22.2)	0 (0.0)
web site with Cochrane summaries	9 (36.0)	1 (12.5)	8 (47.1)	1 (50.0)	0 (0.0)	3 (42.9)	2 (22.2)	3 (75.0)
PubMed or other medical database	14 (56.0)	4 (50.0)	10 (58.8)	2 (100.0)	0 (0.0)	5 (71.4)	7 (77.8)	0 (0.0)
Internet portal about health	4 (16.0)	3 (37.5)	1 (5.9)	0 (0.0)	2 (66.7)	1 (14.3)	1 (11.1)	0 (0.0)
Facebook	4 (16.0)	1 (12.5)	3 (17.6)	1 (50.0)	0 (0.0)	1 (14.3)	2 (22.2)	0 (0.0)
other social network	1 (4.0)	0 (0.0)	1 (5.9)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0)
other	1 (4.0)	0 (0.0)	1 (5.9)	0 (0.0)	1 (33.3)	0 (0.0)	0 (0.0)	0 (0.0)
Did you ever hear about The Cochrane Library?	945	465	480	188	188	191	186	192
yes	24 (2.5)	8 (1.7)	16 (3.3)	2 (1.1)	3 (1.6)	7 (3.7)	8 (4.3)	4 (2.1)
no	921 (97.5)	457 (98.3)	464 (96.7)	186 (98.9)	185 (98.4)	184 (96.3)	178 (95.7)	188 (97.9)
Where did you hear about The Cochrane Library? $^{^{\dagger}}$	23	7	16	2	2	7	8	4
books	4 (17.4)	1 (14.3)	3 (18.6)	0 (0.0)	0 (0.0)	3 (42.9)	1 (12.5)	0 (0.0)
friends and acquaintances that are health care workers	11 (47.8)	3 (42.9)	8 (50.0)	1 (50.0)	0 (0.0)	5 (71.4)	1 (12.5)	4 (100.0)
research manuscripts	10 (43.5)	2 (28.6)	8 (50.0)	0 (0.0)	1 (50.0)	5 (71.4)	3 (37.5)	1 (25.0)
promotional materials of pharmaceutical companies	2 (8.7)	0 (0.0)	2 (12.5)	0 (0.0)	0 (0.0)	1 (14.3)	1 (12.5)	0 (0.0)
Internet	9 (39.1)	3 (42.9)	6 (37.5)	1 (50.0)	2 (100.0)	3 (42.9)	3 (37.5)	0 (0.0)
specialized magazines about medicine for patients	1 (4.3)	0 (0.0)	1 (6.3)	0 (0.0)	0 (0.0)	1 (14.3)	0 (0.0)	0 (0.0)
other	3 (13.0)	1 (14.3)	2 (12.5)	0 (0.0)	0 (0.0)	1 (14.3)	2 (25.0)	0 (0.0)
Do you ever use Cochrane Library?	25	9	16	2	3	7	9	4
yes	13 (52.0)	5 (55.6)	8 (50.0)	2 (100.0)	0 (0.0)	5 (71.4)	6 (66.7)	0 (0.0)
no	12 (48.0)	4 (44.4)	8 (50.0)	0 (0.0)	3 (100.0)	2 (28.6)	3 (33.3)	4 (100.0)
How do you access Cochrane Library? ⁺	12	4	8	2	0	5	5	0
work computer	6 (50.0)	2 (50.0)	4 (50.0)	2 (100.0)	0 (0.0)	2 (40.0)	2 (40.0)	0 (0.0)
home computer	10 (83.3)	2 (50.0)	8 (100.0) 0 (0.0)	0 (0.0)	5 (100.0) 5 (100.0)	0 (0.0)
mobile phone	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)
other	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)

FABLE 3. CONTINUED.	Use of evidence-based	d medicine information sources
---------------------	-----------------------	--------------------------------

	No. (%) of participants								
Variables	Entire sample (N=978)	Rural areas (n=482)	Urban areas (n=496)	Osijek region (n=189)	Rijeka region (n=193)	Split region (n=200)	Varaždin region (n=199)	Zagreb region (n=197)	
How often do you use Cochrane Library?	12	4	8	2	0	5	5	0	
less than once a year	1 (8.3)	0 (0.0)	1 (12.5)	0 (0.0)	0 (0.0)	1 (20.0)	0 (0.0)	0 (0.0)	
once a year	3 (25.0)	1 (25.0)	2 (25.0)	0 (0.0)	0 (0.0)	1 (20.0)	2 (40.0)	0 (0.0)	
once in 6 months	3 (25.0)	2 (50.0)	1 (12.5)	1 (50.0)	0 (0.0)	2 (40.0)	0 (0.0)	0 (0.0)	
once a week	4 (33.4)	1 (25.0)	3 (37.5)	1 (50.0)	0 (0.0)	1 (20.0)	2 (40.0)	0 (0.0)	
more times a week	1 (8.3)	0 (0.0)	1 (12.5)	0 (0.0)	0 (0.0)	0 (0.0)	1 (20.0)	0 (0.0)	
How useful did you find information from Cochrane Library?	11	3	8	2	0	5	4	0	
useless	1 (9.1)	0 (0.0)	1 (12.5)	0 (0.0)	0 (0.0)	1 (20.0)	0 (0.0)	0 (0.0)	
undecided	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	
useful	10 (90.9)	3 (100.0)	7 (87.5)	2 (100.0)	0 (0.0)	4 (80.0)	4 (100.0)	0 (0.0)	
v^{2} test (P < 0.05).									

[†]Multiple choice guestion.

indicated they heard of Cochrane, majority indicated that they came across it on PubMed or other medical database or the Cochrane summaries website. There were 3% of patients who indicated that they heard of The Cochrane Library, mostly from friends and acquaintances who are health care workers, from a research paper, and the Internet. Of 25 patients who indicated they had heard of The Cochrane Library, 13 indicated that they used it and mostly accessed it from a computer at work. The Cochrane Library was used rarely and most participants did found it useful (Table 3).

In our sample, there was a significant difference between patients from urban and rural areas in the number of patients-students, patients with primary-level education, and university-level education (χ^2 test, P < 0.001 for all), whether they searched for information on diagnosis or therapy other than with physician (χ^2 test, P = 0.012), and how often they used the Internet (χ^2 test, P < 0.001). When different regions from our sample were compared, there was a significant difference between the number of employed patients (the lowest number in the Rijeka region; χ^2 test, P < 0.001), number of retired patients (the lowest number in Varaždin region; χ^2 test, P < 0.001), and in daily use of Internet (the highest number in Varaždin region; χ^2 test, P < 0.001) (Tables 1 and 2).

Higher awareness of EBM was associated with the urban place of residence (χ^2 test, P < 0.001), higher level of education (χ^2 test, P < 0.001), and employment status (χ^2 test, P = 0.001).

Patients from Osijek region were most familiar with EBM compared with other regions (χ^2 test, P < 0.001). Awareness about systematic reviews was associated with higher level of education (χ^2 test, P < 0.001) and employment status (χ^2 test, P < 0.001). In comparison with other regions, patients from Varaždin region were most familiar with systematic review term (χ^2 test, P = 0.002).

Among 181 patients who heard about EBM, there were only 61 (34%) patients who heard about both EBM and systematic review. Among 160 participants who heard about a systematic review, 99 (61.9%) did not hear about EBM (χ^2 test, P < 0.001). Using the logistic regression model, we showed that the educational level and urban/rural residence were predictors for awareness about EBM and awareness about systematic reviews (P < 0.001 for both).

When asked to describe in their own words the meaning of EBM, 129 participants responded (Supplementary Table 1). Majority (n = 102) answered that it was medicine based on research, verified medicine, tested on a large number of patients. The second most common group of answers indicated that EBM is a Western, or European, or non-alternative, recognized medicine (n = 5). Many patients equated EBM with the use of certain medical therapies, and one patient answered that EBM is "the TV show of Dr. Oz".

Only 3% of patients heard of the Cochrane Croatia and 2% visited 'Cochrane Health' Facebook page. When asked if they read translated Cochrane PLSs, 2% of patients gave a confirmatory answer. When ranking useful-

305

ness of various information sources, physician's opinion was ranked the highest, followed by systematic reviews, research conducted on humans and experiences of other patients, while information from media were ranked the lowest (Table 4).

When asked whether they ever discuss medical information they found on their own with their physician, 245 patients confirmed it. Majority said that the physician's reaction was positive or good, and only a few patients reported a negative reaction from a physician (Supplementary Table 2).

DISCUSSION

We found that half of the patients searched for medical information from sources other than physician and mostly on the Internet. Very few patients used EBM sources of medical information; one-fifth of patients heard of the EBM, but very few heard of the Cochrane. Patients considered the physician's opinion as the most reliable source of medical information. Place of residence and educational level were predictors of awareness about EBM and systematic reviews.

Health information seeking is crucial for today's participatory role of patients in health care (18). Information seeking is noted as a critical component of shared decision-making because informed patients will more likely participate in choosing their care, make wiser decisions, and adhere more fully to the treatment (19). This study addressed general health information seeking habits, with a special focus on the Cochrane evidence as the gold standard in EBM information. Since Cochrane Croatia was founded, a significant volunteer effort was devoted to translating PLSs into Croatian language and their promotion via social media (20). Our results indicate that additional strategies need to be considered to reach a wider consumer audience in order to disseminate Cochrane evidence, but keeping focus on dissemination through web and social media is also relevant.

Although social media are relatively new, they are already used in health care. There are reports about different patterns of social media use between patients and health care professionals, indicating that patients prefer using Facebook for social support and exchanging advice (21). Web has a huge potential as a resource for high-quality public health information. However, since there are various forms of information, this can lead to unintended consequences, such as misinformation and patient fraud (22). Therefore, it is important to promote high-quality, non-commercial EBM information sources among patients.

TABLE 4. Participants' awareness of Cochrane Croatia activities

	No. (%) of participants							
Variables	Entire sample (N=978)	Rural areas (n=482)	Urban areas (n=496)	Osijek region (n = 189)	Rijeka region (n = 193)	Split region (n=200)	Varaždin region (n = 199)	Zagre region (n = 197)
Did you ever hear about Cochrane Croatia?	927	462	465	186	180	190	180	191
yes	31 (3.3)	11 (2.4)	20 (4.3)	4 (2.2)	3 (1.7)	8 (4.2)	10 (5.6)	6 (3.1)
no	896 (96.7)451 (97.6)	445 (95.7)182 (97.8)	177 (98.3)	182 (95.8	170 (94.4)	185 (96.9)
Did you ever visit a 'Cochrane health' page on Facebook?	928	463	465	187	183	187	178	193
yes	15 (1.6)	5 (1.1)	10 (2.2)	2 (1.1)	3 (1.6)	3 (1.6)	5 (2.8)	2 (1.0)
no	913 (98.4)	458 (98.9)455 (97.8)	185 (98.9)	180 (98.4)	184 (98.4)173 (97.2)	191 (99.0)
Did you ever read translated plain language summaries in Croatian language prepared by the Cochrane Croatia?	934	468	466	187	182	188	183	194
yes	22 (2.4)	7 (1.5)	15 (3.2)	4 (2.1)	3 (1.6)	5 (2.7)	7 (3.8)	3 (1.5)
no	912 (97.6)	461 (98.5)451 (96.8)	183 (97.9)	179 (98.4)	183 (97.3)	176 (96.2)	191 (98.5)
Please rate these medical information sources based on your perception of their reliability (median, interquartile range)								
experiences of other patients	3 (2-4)	3 (3-4)	3 (2-4)	2 (2-3)	3 (2-4)	3 (3-4)	3 (2-3)	4 (3-4)
research conducted on humans	4 (3-5)	4 (3-5)	4 (3-4)	3 (3-4)	4 (3-4)	5 (4-5)	4 (3-4)	4 (3-5)
analysis of multiple studies (a systematic review)	4 (3-5)	4 (4-5)	4 (3-5)	3 (3-4)	4 (3-5)	5 (4-5)	4 (3-4)	4 (4-5)
information from media	3 (2-3)	3 (2-4)	2 (2-3)	2 (1-3)	3 (2-4)	3 (2-3)	3 (2-3)	2 (2-3)
physician's opinion	5 (4-5)	5 (4-5)	5 (4-5)	5 (5-5)	5 (4-5)	5 (4-5)	4 (4-5)	5 (4-5)

Cochrane Reviews have considerably expanded their reach and the access to the Cochrane Database of Systematic Reviews, where the reviews are published, has been increasing continuously (23). Since Cochrane Reviews are prepared and published in English, the Cochrane Collaboration has devoted a considerable effort to translating the evidence for use by patients, ie, the public. Different strategies could be used for targeted dissemination to specific patient groups (24). Translating summaries of Cochrane into multiple languages is an important part of the Cochrane strategy titled "Evidence to everyone, everywhere" (23). At the moment, Croatian is one of thirteen languages to which Cochrane PLSs are being translated.

In this study, most respondents defined EBM as a practice of medicine justified by research results. It was curious that one person wrote that EBM was the Dr Oz Show. Contents of this TV show were recently scrutinized by researchers. Their research concluded that consumers should be very skeptical about recommendations provided on medical talk shows, as only a third to one-half of the recommendations given was based on believable or somewhat believable evidence (25).

When asked to rank different sources of medical information, patients indicated physician's opinion as the most reliable, followed by a systematic review, as an analysis of multiple studies. Information from the media was ranked the lowest. According to the EBM principles, systematic reviews are the highest level in the hierarchy of evidence in medicine (26), whereas expert opinion is ranked much lower, even lower than observational studies (27). However, that does not mean that clinical expertise is irrelevant in EBM. On the contrary, the practice of EBM involves integrating the appraisal of evidence with clinical expertise and patient values to apply the results in clinical practice (26). Our findings that patients consider physician's opinion as the most reliable source of medical information could be used for knowledge translation and physicians need to be enticed to promote EBM information sources, such as The Cochrane Library and translated Cochrane evidence, among patients. Curricula of medical specialty training for physicians then also need to include EBM and knowledge translation skills, if they are not already included. Since we found that the predictors of awareness about EBM and systematic reviews were the place of residence and educational level, more effort should be devoted to education of patients in rural areas and those with less education.

A limitation of this study may be patient sampling, which was convenient and, therefore, can be considered nonrepresentative of five large geographic areas. However, we made an effort to include patients from both urban and rural areas, from as much as 10 different family medicine practices. The sample was not entirely balanced, as in rural areas more participants had lower educational level. Second, the sample was drawn from patients visiting a physician. Third, it is possible that participants gave positive answers to guestions about EBM and systematic review awareness even though they were not familiar with these concepts. Still, very few patients responded positively to those questions and responses were uniform between regions. Other limitations may be related to authors' professional activities, as several study authors are personally involved in the activities of Cochrane Croatia and have a track record of publications on EBM. However, only one of them participated in data collection, without giving any instructions about the manuscript to the recruited participants. To ensure that we were not biased when creating our questionnaire, it has been piloted and validated among researchers, students, and lay persons. Finally, there is potentially an upward bias present in our results.

While a number of studies about knowledge, awareness, and attitudes about EBM were conducted among various types of health care workers in different settings worldwide (17,28-31), we believe this is the first study that investigates awareness and use of EBM information among patients on a nation-wide scale, which is a strength of this study.

Even though half of the included patients used Internet in their search for medical information, very few were familiar with EBM, sources such as the Cochrane and few used these sources. Place of residence and education level were predictors of awareness about EBM and systematic reviews. Our finding that patients find physician's opinion to be the most reliable source of health-related information could be used for promotion of high-quality health information among patients and an incentive to train physicians in EBM and knowledge translation skills.

Acknowledgment We are grateful to colleagues from Cochrane Croatia for providing helpful comments related to the study questionnaire during development of the study protocol, including Irena Zakarija Grković, Vanja Giljača, Mario Malički and Davor Lukšić. Also, we wish to thank Katarina Madirazza for her help with language editing of the manuscript. Many thanks to Mario Malički for his generous statistical advice.

Funding None.

Ethical approval received from the Ethics Committee of the University of Split School of Medicine (Class: 003-08/14-03/0001; Reg.No.: 2181-198-03-04-14-033).

307

Declaration of authorship DN, LP, and DV conceived the study and designed the questionnaire for the survey. IM, DV, SPH, MT, RG, IDL, JSH, VP, VC, AT, SB, MJ, and KT substantially contributed to acquisition of data. DN and LP analyzed the data. DN and LP drafted the first version of the manuscript. All authors were involved in the interpretation of research findings, critical revision of the manuscript, and final approval of the manuscript. All authors agree to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Competing interests All authors have completed the Unified Competing Interest form at www.icmje.org/coi_disclosure.pdf (available on request from the corresponding author) and declare: no support from any organization for the submitted work; no financial relationships with any organizations that might have an interest in the submitted work in the previous 3 years; no other relationships or activities that could appear to have influenced the submitted work.

References

- Pavličević I. Patient opinion is important: patients should decide on their treatment with doctors [in Croatian]. Medix. 2014;20:116-9.
- 2 The Cochrane Collaboration. Trusted evidence. Informed decisions. Better health. 2015. Available from: http://www.cochrane.org. Accessed: August 03, 2015.
- 3 Institute of Medicine Committee on Quality of Health Care in A. Crossing the quality chasm: a new health system for the 21st century. Washington (DC): National Academies Press (US); 2001.
- 4 Simi S, Patoia L. The importance of being patient. J Ambul Care Manage. 2010;33:257-64. Medline:20539153 doi:10.1097/ JAC.0b013e3181e62bf3
- 5 Stewart M, Brown J, Weston W, McWhinney I, McWilliam C, Freeman T. Patient-centered medicine: transforming the clinical method. Thousand Oaks (CA): Sage Publications; 1995.
- 6 Sullivan M. The new subjective medicine: taking the patient's point of view on health care and health. Soc Sci Med. 2003;56:1595-604. Medline:12614708 doi:10.1016/S0277-9536(02)00159-4
- 7 DeMaria AN. The informed patient. J Am Coll Cardiol. 2008;51:1505-6. Medline:18402907 doi:10.1016/j.jacc.2008.03.013
- 8 Self-Diagnosis poll executive summary [white paper]. Wolters Kluwer Health; 2012. Available from: http://www. wolterskluwerhealth.com/News/Documents/White%20Papers/ Self-Diagnois%20Poll.pdf. Accessed: August 03, 2015.
- 9 Fox S. Pew Internet & American Life Project. Health online 2013; 2013. Available from: http://www.pewinternet.org/files/oldmedia//Files/Reports/PIP_HealthOnline.pdf. Accessed: August 03, 2015.
- 10 Andreassen HK, Bujnowska-Fedak MM, Chronaki CE, Dumitru RC, Pudule I, Santana S, et al. European citizens' use of E-health services: a study of seven countries. BMC Public Health. 2007;7:53. Medline:17425798 doi:10.1186/1471-2458-7-53
- Sackett DL, Rosenberg WM, Gray JA, Haynes RB, Richardson WS. Evidence based medicine: what it is and what it isn't. BMJ. 1996;312:71-2. Medline:8555924 doi:10.1136/bmj.312.7023.71
- 12 PubMed U.S. National Library of Medicine [Internet]. 2015. Available from: http://www.ncbi.nlm.nih.gov/pubmed. Accessed:

August 03, 2015.

- 13 The Cochrane Library 2015. Available from: http://www. cochranelibrary.com. Accessed: August 03, 2015.
- 14 Puljak L, Rako D. Enhancing medical practice in Croatia through The Cochrane Collaboration. Biochem Med. 2009;19:260-5. doi:10.11613/BM.2009.024
- The Cochrane Croatia Croatian Branch of Italian Cochrane Centre.
 2015. Available from: http://croatia.cochrane.org. Accessed: August 03, 2015.
- 16 Facebook page of Cochrane Croatia. Available from: https://www.facebook.com/pages/Hrvatski-Cochraneogranak/163424700478175. Accessed: August 03, 2015.
- 17 Novak K, Mirić D, Jurin A, Vukojević K, Aljinović J, Carić A, et al. Awareness and use of evidence-based medicine databases and Cochrane Library among physicians in Croatia. Croat Med J. 2010;51:157-64. Medline:20401959 doi:10.3325/cmj.2010.51.157
- Anker AE, Reinhart AM, Feeley TH. Health information seeking: a review of measures and methods. Patient Educ Couns.
 2011;82:346-54. Medline:21239134 doi:10.1016/j.pec.2010.12.008
- Epstein RM, Alper BS, Quill TE. Communicating evidence for participatory decision making. JAMA. 2004;291:2359-66.
 Medline:15150208 doi:10.1001/jama.291.19.2359
- 20 Puljak L. Using social media for knowledge translation, promotion of evidence-based medicine and high-quality information on health. J Evid Based Med. 2016;9:4-7. Medline:26372327 doi:10.1111/jebm.12175
- 21 Antheunis ML, Tates K, Nieboer TE. Patients' and health professionals' use of social media in health care: motives, barriers and expectations. Patient Educ Couns. 2013;92:426-31. Medline:23899831 doi:10.1016/j.pec.2013.06.020
- 22 Kelly L, Ziebland S, Jenkinson C. Measuring the effects of online health information: Scale validation for the e-Health Impact Questionnaire. Patient Educ Couns. 2015;98:1418-24. Medline:26162953 doi:10.1016/j.pec.2015.06.008
- 23 von Elm E, Ravaud P, Maclehose H, Mbuagbaw L, Garner P, Ried J, et al. Translating Cochrane reviews to ensure that healthcare decision-making is informed by high-quality research evidence. PLoS Med. 2013;10:e1001516. Medline:24068897 doi:10.1371/ journal.pmed.1001516
- Santesso N, Glenton C, Lang B. Evidence that patients can understand and use? Z Evid Fortbild Qual Gesundhwes.
 2008;102:493-6. Medline:19216198 doi:10.1016/j.zefq.2008.08.027
- 25 Korownyk C, Kolber MR, McCormack J, Lam V, Overbo K, Cotton C, et al. Televised medical talk shows–what they recommend and the evidence to support their recommendations: a prospective observational study. BMJ. 2014;349:g7346. Medline:25520234 doi:10.1136/bmj.g7346
- 26 Straus SE, McAlister FA. Evidence-based medicine: a commentary on common criticisms. CMAJ. 2000;163:837-41. Medline:11033714
- 27 Brighton B, Bhandari M, Tornetta P III, Felson DT. Hierarchy of

evidence: from case reports to randomized controlled trials. Clin Orthop Relat Res. 2003;19-24. Medline:12897592 doi:10.1097/01. blo.0000079323.41006.12

- 28 Jette DU, Bacon K, Batty C, Carlson M, Ferland A, Hemingway RD, et al. Evidence-based practice: beliefs, attitudes, knowledge, and behaviors of physical therapists. Phys Ther. 2003;83:786-805. Medline:12940766
- 29 Amin FA, Fedorowicz Z, Montgomery AJ. A study of knowledge and attitudes towards the use of evidence-based medicine among primary health care physicians in Bahrain. Saudi Med J. 2006;27:1394-6. Medline:16951780
- Egerod I, Hansen GM. Evidence-based practice among Danish cardiac nurses: a national survey. J Adv Nurs. 2005;51:465-73.
 Medline:16098163 doi:10.1111/j.1365-2648.2005.03525.x
- 31 Gray M, Joy E, Plath D, Webb SA. Opinions about evidence: a study of social workers' attitudes towards evidencebased practice. Journal of Social Work. 2014;14:23-40. doi:10.1177/1468017313475555